## RESEARCH

**Open Access** 



# Known HIV status among adolescent women attending antenatal care services in Eswatini, Ethiopia and Mozambique

Joel Njah<sup>1</sup>, Mary Ann Chiasson<sup>2,3</sup> and William Reidy<sup>1,2\*</sup>

## Abstract

**Background:** Antenatal care (ANC) clinics remain important entry points to HIV care for pregnant women living with HIV—including adolescents. Prior knowledge of their HIV status at ANC enrollment is crucial to providing services for prevention of mother-to-child transmission (PMTCT) of HIV. We examined known HIV status of pregnant adolescents and women in other age groups at ANC enrollment.

**Methods:** A descriptive study of routinely reported PMTCT data from 419 facilities in Eswatini, Ethiopia, and Mozambique, from January through December 2018 was conducted. We assessed knowledge of HIV status by country for three age groups: adolescents aged 15–19 years, young women aged 20–24 years, and older women aged 25–49 years. We report HIV prevalence and proportions of known and newly diagnosed HIV infections in women, by age group and country. The data were summarized by frequencies and proportions, including their 95% confidence intervals.

**Results:** Among the facilities examined, 52 (12.4%) were in Eswatini, 63 (15.0%) in Ethiopia, and 304 (72.6%) in Mozambique. Across three countries, 488,121 women attended a first ANC visit and 23,917 (4.9%) were HIV-positive. Adolescents constituted 22% of all ANC attendees, whereas young and older women represented 33% and 45%, respectively. HIV prevalence was lowest among adolescents than in other age groups in Eswatini (adolescents 11.9%, young 24.2% and older 47.3%), but comparable to young women in Ethiopia (adolescents 1.6%, young 1.6% and older 2.2%) and Mozambique (adolescents 2.5%, young 2.5% and older 5.8%), However, in each of the three countries, lower proportions of adolescents knew their HIV-positive status before ANC enrollment compared to other age groups: in Eswatini (adolescents 51.3%, young 59.9% and older 79.2%), in Ethiopia (adolescents 42.9%, young 63.7% and older 75.2%), and in Mozambique (adolescents 16.4%, young 33.2% and older 45.6%).

**Conclusion:** Overall, adolescents made up nearly one-quarter of the women examined and had the least knowledge of their HIV status at ANC enrollment. Their HIV prevalence and known HIV-positive status varied widely across the countries examined. Adolescent-friendly sexual and reproductive health, and PMTCT services, before pregnancy, are needed to improve knowledge of HIV status and support pregnant adolescents and their infants.

**Plain English summary:** Antenatal care (ANC) clinics are important for HIV testing of pregnant adolescents, who may not know their HIV-positive status at the first ANC visit. We describe data on pregnant adolescents and women in other age groups in ANC services to examine their prior HIV status at ANC enrollment across three African countries.

168th St., New York, NY 10032, USA

Full list of author information is available at the end of the article



© The Author(s) 2021. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/ficenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

<sup>\*</sup>Correspondence: wr2205@cumc.columbia.edu

 $<sup>^{\</sup>rm 1}$  ICAP At Columbia University's Mailman School of Public Health, 722 W.

We examined data from 419 PMTCT sites in Eswatini, Ethiopia, and Mozambique from January-December 2018, to evaluate HIV testing results for adolescents, young and older women aged 15–19, 20–24 and 25–49 years, respectively. We report the number of women living with HIV and the proportions of known and newly identified women living with HIV, by age-group and country.

Across three countries, 488,121 women attended ANC and 23,917 (4.9%) were living with HIV. Adolescents constituted 22% of all ANC attendees, whereas young and older women represented 33% and 45%, respectively. HIV prevalence in each country compared to other age groups was lowest and varied among adolescents from 11.9% in Eswatini, to 1.6% in Ethiopia and to 2.5% in Mozambique. Also, fewer adolescents knew their HIV-positive status before ANC enrollment compared to young and older women from 51.3% in Eswatini, 42.9% in Ethiopia to only 16.4% in Mozambique.

Pregnant adolescents made up nearly one-quarter of all ANC attendees; a majority of them had no previously known HIV-positive status. Adolescent-friendly, sexual and reproductive health services, before pregnancy and in PMTCT services, are needed to support pregnant adolescents and their infants.

Keywords: Adolescent pregnancies, HIV testing results, ANC, PMTCT, Sub-saharan Africa

## Background

Over the past decade, remarkable progress has been made towards eliminating perinatal transmission of HIV through the global scale-up of prevention of mother-tochild transmission (PMTCT) of HIV services, particularly in sub-Saharan Africa (SSA) [1]. However, the risk of acquiring HIV infection is disproportionately higher among adolescent girls who together with young women under age 25 account for 30% of new HIV infections, though they make up just 10% of the population [2, 3]. HIV/AIDS was reported as the second leading cause of death in South Africa in 2015 among adolescent girls and young women (AGYW) [4], and the trend in mortality is projected to increase among this age group [5]. In response, a global target of fewer than 100,000 new HIV infections among AGYW aged 15 to 24 years by 2020 was established [2]. Recent reports, however, indicate that this target has been missed because of the slow decline in new HIV infections among AGYW [6, 7]. The slow decline might be due to the increased vulnerability of adolescent girls to HIV infection that is compounded by high rates of adolescent pregnancies in SSA. To shed light on the high prevalence of adolescent pregnancies, a United Nations Population Fund (UNFPA) global review, found that more than 25% of young women, including adolescents, examined had been pregnant by age 18 [8]. Exceptionally high pregnancy rates were reported in the Eastern and Southern regions of Africa, in particular in Mozambique, where over 40% of adolescent girls had at least one pregnancy [8].

Given the high pregnancy rates and disproportionately high HIV incidence among adolescent girls in SSA, substantial numbers of adolescent girls living with HIV (ALHIV) will be in need of antenatal care (ANC) and PMTCT services. ANC services serve as critical entry points for HIV testing and linkage to PMTCT care [9]; however, previous data from Zimbabwe have shown that only 3.1% of adolescents had a knowledge of their HIV status at ANC enrollment [10]. This is important because knowledge of HIV status prior to or during a first ANC visit may impact the quality and long-term engagement of adolescents in PMTCT services. In South Africa for example, a high LTFU rate of 57.5% was observed among newly identified women living with HIV (WLHIV), initiating antiretroviral therapy (ART) at a first ANC visit in a major urban clinic [11]. Looking specifically at adolescents initiating ART in three Eastern African countries, a recent publication found that pregnant adolescents aged 15–19 years had a higher risk of LTFU from HIV care when compared to their non-pregnant female counterparts [12].

Collectively, the extant literature reveals gaps in knowledge of HIV status among pregnant adolescents before ANC enrollment. We sought to analyze routine facilitylevel data from PMTCT sites in three SSA countries to understand gaps in HIV status knowledge among pregnant women of different age groups at a first ANC visit.

#### Methods

#### Study setting and design

This analysis used routine reporting data from PMTCT facilities in Eswatini, Ethiopia and Mozambique supported by ICAP at Columbia University through funding from the US President's Emergency Plan for AIDS Relief (PEPFAR). Facility-level, age-disaggregated PMTCT program data were collected from January through December 2018. Of a total of 419 ICAP-supported facilities, 52 sites are in Eswatini; a Southern African country with four administrative regions, including Manzini where all the 52 sites are located. This landlocked country has a population of about 1.3 million inhabitants with an HIV prevalence of 27%; the highest in the world [13, 14]. Sixty

three (63) of the ICAP-supported sites are located in four of the 9 administrative regions of Ethiopia, which are Afar, Somali, Benishagul-Gumuz and Gambella. Ethiopia is an Eastern African country with over 114 million inhabitants and a generalized HIV prevalence of 0.9% [15]. The bulk of the data for this analysis were obtained from 304 ICAP-supported sites in Mozambique. Mozambique is a Southern African country with more than 29 million inhabitants distributed in ten administrative provinces with a national HIV prevalence of 12.4%. These ICAP-supported sites are located in the two most populous provinces of Nampula and Zambezia. Nampula has an HIV prevalence of 6.5% compared to 14.5% in Zambezia.

### Data source and outcome variables

Data for pregnant women receiving PMTCT services were routinely collected by facility health personnel and aggregated at the facility level for reporting to national ministries of health and to Centers for Disease Control and Prevention (CDC)/PEPFAR. The data on numbers of first ANC attendees, previous knowledge of HIVpositive status and HIV-positive diagnosis in ANC were examined for three age groups that included adolescents aged 15-19 years, young women aged 20-24 years, and older women aged 25-49 years. Younger adolescents 10–14 years old were excluded from the analysis because of lack of disaggregated data on 'known' or 'newly' diagnosed as living with HIV that were needed to calculate the proportions. The outcome variables were the agestratified prevalence of WLHIV, comprising those with known HIV-positive status prior to ANC enrollment plus those who newly diagnosed as living with HIV at ANC enrollment among all ANC clients; and the proportion of women previously known to be living with HIV among all women diagnosed as living with HIV were assessed across three countries [16]. Across ICAP-supported facilities, known HIV-positive status was confirmed for clients not on antiretroviral therapy (ART) and for those newly diagnosed as living with HIV, positive HIV-status was established by conducting two serial rapid HIV diagnostic tests according to WHO guidelines [17]. A commonly used first rapid test is the Abbott Determine<sup>™</sup> HIV-1/2 Ag/Ab strips (Abbott Laboratories, Abbott Park, Illinois, USA), but the choice of the serial rapid tests varied by country [18].

#### Data analysis

Our findings were described and summarized by frequencies and proportions, including their 95% confidence intervals (CI) and graphs of proportions by age groups for each country. We also conducted a Kruskal Wallis test to compare the mean ranks of the prevalence among women with known HIV status by country and age group. The age-disaggregated data were downloaded from the ICAP central database as excel sheets and imported into STATA version 15.0 (StataCorp, College Station, TX) for analysis.

#### Results

## Study health facilities

Overall, 419 ICAP-supported PMTCT facilities were examined, including 52 (12.4%) in the Manzini region of Eswatini, 63 (15.0%) in Afar, Benishangul-Gumuz, Gambella and Somali regions of Ethiopia, and 304 (72.6%) in Nampula and Zambezia provinces of Mozambique.

(Fig. 1).

#### **Study population**

A total of 488,121 pregnant women attended a first ANC visit between January and December 2018 across the 419 health facilities in the three countries. Of these, 483,246 (99.0%) women aged 15–49 years were included in the analysis; 4,875 women who were either younger than 15 years or older than 49 years were excluded from the analysis. Among those included in the analysis, 23,917 (4.9%) were living with HIV (Table 1).

## HIV prevalence among all women attending a first ANC visit

Overall, across three countries, adolescents constituted 22.0% (106,531/483,248) of all women seeking a first ANC visit, while 32.9% (159,210/483,248) and 45.0% (217,507/483,248) of them were young and older women, respectively. By country, HIV prevalence was lowest among adolescents in Eswatini (adolescents 11.9% [95% CI: 10.2-13.6], young 24.2% [95% CI: 20.5-27.9], and older 47.3% [95% CI: 45.8-48.9]), but similar to that of young women in Ethiopia (adolescents 1.6% [95% CI: 1.0-2.3], young 1.6% [95% CI: 1.0-2.1] and older 2.2% [95% CI: 1.4-3.0]), and Mozambique (adolescents 2.5% [95% CI: 2.2-2.9], young 2.5% [95% CI: 3.6-5.5], and older 5.8% [95% CI: 5.0-6.5]) (Fig. 2a). The prevalence of HIV-positive women and proportion of women with previous knowledge of their HIV-positive status at ANC enrollment bay age group. a Prevalence of all HIV-positive women by age group among first visit ANC attendees. b Percentage of women with prior knowledge of their HIV-positive status and those newly tested HIV-positive among all HIV-positive women at ANC enrollment. The prevalence of HIV-positive women and proportion of women with previous knowledge of their HIV-positive status at ANC enrollment bay age group. a Prevalence of all HIV-positive women by age group among first visit ANC attendees. b Percentage of women with prior knowledge of their HIV-positive status and those newly



**Table 1** Number of new ANC clients, prevalence and proportions of women with previously known HIV-positive status at ANC enrollment by age group and country in 2018

	Age group (years)	Eswatini, N = 8509 (%)	Ethiopia, N = 43,893 (%)	Mozambique, N = 435,719 (%)
ANC clients by age group	15–19	1313	6524	98,694
	20–24	2497	15,081	141,632
	25–49	4678	22,254	190,575
All HIV-positive women among new ANC clients <sup>a</sup>		2975 (35.0)	822 (1.9)	20,120 (4.6)
All HIV-positive women among new ANC clients by age group <sup>a,b</sup>	15–19	156 (11.9)	105 (1.6)	2490 (2.5)
	20–24	604 (24.2)	234 (1.6)	6497 (2.5)
	25–49	2212 (47.3)	483 (2.2)	11,100 (5.8)
Women with previously known HIV status among all HIV-positive women by age group <sup>c</sup>	15–19	80 (51.3)	45 (42.9)	408 (16.4)
	20–24	362 (59.9)	149 (63.7)	2156 (33.2)
	25–49	1751 (79.2)	363 (75.2)	5065 (45.6)

ANC antenatal care

<sup>a</sup> Overall prevalence of HIV-positive pregnant women

 $^{\rm b}$  Prevalence of HIV-positive pregnant women by age group

<sup>c</sup> Proportion of pregnant women with previously known HIV-positive status before ANC enrollment by age group

tested HIV-positive among all HIV-positive women at ANC enrollment. The prevalence of HIV-positive women and proportion of women with previous knowledge of their HIV-positive status at ANC enrollment bay age group. a Prevalence of all HIV-positive women by age group among first visit ANC attendees. b Percentage of women with prior knowledge of their HIV-positive status and those newly tested HIV-positive among all HIV-positive women at ANC enrollment.

## Proportions of women known to be living with HIV at ANC enrollment

Across three countries, adolescent women had the lowest proportion of women known to be living with HIV



before ANC enrollment at 11.5%, whereas young and older women represented 30.7% and 57.8%, respectively. The proportion of all WLHIV with known HIV positive status, as opposed to those newly diagnosed at ANC enrollment, across all age groups was highest in Eswatini at 73.7%, followed by 67.8% in Ethiopia and 37.9% in Mozambique. Across three countries, smaller proportions of adolescent pregnant women knew their HIV-positive status before ANC enrollment; in Eswatini (adolescents 51.3% [95% CI: 27.8-77.6], young 59.9% [95% CI: 45.0-76.7] and older 79.2% [95% CI: 75.0-83.2]), Ethiopia (adolescents 42.9% [95% CI: 27.9-60.2], young 63.7% [95% CI: 47.3-80.0] and older 75.2% [95% CI: 69.0-82.6]), and Mozambique (adolescents 16.4% [95% CI: 11.4–21.1], young 33.2% [95% CI: 26.2–39.0] and older 45.6% [95% CI: 40.1–50.5]) (Fig. 2b).

Comparing the HIV prevalence among women with known HIV status by countries and age groups, there was a statistically significant increase among adolescent women between: Eswatini and Mozambique (p-value=0.004), Ethiopia and Mozambique (p-value=0.02), but not between Eswatini and Ethiopia (p-value=0.278). A similar trend was

observed among young women, Eswatini and Mozambique (p-value=0.016), Ethiopia and Mozambique (p-value=0.005), but not between Eswatini and Ethiopia (p-value=0.347), and among older women, Eswatini and Mozambique (p-value=0.004), Ethiopia and Mozambique (p-value=0.02), but not between Eswatini and Ethiopia (p-value=0.278).

#### Discussion

In this analysis, we found that nearly a quarter (22.0%) of the women seeking ANC services were adolescent aged 15–19 years and that the proportions of adolescent women with known HIV-positive status before ANC enrollment varied considerably by country, ranging from 51% in Eswatini to 43% in Ethiopia to only 16% in Mozambique. We also found that lower proportions of pregnant adolescent knew their HIV-positive status before ANC enrollment compared to women in the young and older age groups across three countries. This report underscores the need to promote and expand adolescent-friendly HIV testing services to increase the HIV status knowledge of adolescent girls prior to a first ANC visit [19, 20].

Our findings are consistent with previous work by investigators in South Africa, who showed that adolescent women below 20 years of age were two times more likely than older women not to know their HIVpositive status before attending ANC services [21]. Musarandenga et al., assessing the uptake of PMTCT services among women attending ANC services in Zimbabwe, found that only 3% of adolescent women had knowledge of their HIV status at ANC enrollment [10]. Similarly, other studies in South Africa [22], and Kenya [20], showed that 75% and 70% of pregnant adolescents did not know their HIV status before ANC enrollment, respectively.

The existing body of literature, points to a relatively low overall HIV testing coverage among non-pregnant adolescents [10, 23]. For example, only 19% of girls and 14% of boys aged 15-19 in the Eastern and Southern Africa region tested for HIV and got their results in 2018 [24]. Reasons for this low uptake in HIV testing among adolescents according to a recent study include: low perception of HIV infection risk, fear of HIV-related stigma, and discomfort with finding out they are HIV-positive [4]. Testing for HIV and active linkage to care and treatment following a positive test result are critical steps in the HIV care continuum for adolescents. For instance, less than half of the adolescents who tested positive for HIV initiated ART in South Africa according to a recent national laboratory study [25]. There is therefore, a need to identify effective interventions that can improve HIV testing among adolescents before they become pregnant. The DREAMS (Determined, Resilient, Empowered, AIDSfree, and Safe) initiative is one of such adolescent-targeted programs that need to be expanded to priority and key adolescent girls to improve their HIV status knowledge before pregnancy [26–28].

Pregnant adolescents face additional challenges and have been shown to be more vulnerable to LTFU after receiving a positive HIV test result during ANC enrollment [29]. While data are limited on the reasons for this increase vulnerability, it may be related to the extra burden of processing new information on testing positive for HIV and initiating ART on the same day, on top of the routine ANC demands. However, adolescent-friendly services are limited in existing PMTCT services. Targeted-adolescent and youth-friendly services have been suggested as solutions to the challenges faced by adolescents in lifelong HIV care and treatment services [23, 30-32]. There is, therefore, an urgent need for innovative service delivery models for pregnant adolescents in ANC and PMTCT services to improve their health outcomes [33, 34].

#### Limitations

The data used in our analyses have some limitations. The facilities we examined include only a small subset of all ANC facilities within the three countries; thus, our findings are not necessarily generalizable to the respective countries. In addition, other than age, the lack of data on hand on patient-level and facility-level characteristics, reduced our ability to assess associations between potential risk factors of poor knowledge of HIV status and PMTCT health outcomes. Future research would be needed to address this limitation. Despite these limitations, the current study provides additional evidence on gaps to fill on improving knowledge of HIV status among adolescent mothers before ANC enrollment.

#### Conclusion

Our analysis showed that nearly one-quarter of all women living with HIV attending PMTCT services were adolescents, and that a substantial number of them were newly diagnosed with HIV infection at ANC enrollment. Also, lower proportions of adolescents, than women in other age groups, knew their HIV status. Adolescent-friendly sexual and reproductive health, and PMTCT services, before pregnancy, are needed to improve knowledge of HIV status and support these adolescent mothers and their infants.

#### Abbreviations

MTCT: Mother-to-child transmission of HIV; PMTCT: Prevention of mother-tochild transmission of HIV; ANC: Antenatal care; SSA: Sub-Saharan Africa; AGYW : Adolescent girls and young women; LTFU: Loss to follow-up; DREAMS: Determined, Resilient, Empowered, AIDS-free, Mentored, and Safe.; WLHIV: Women living with HIV; ALHIV: Adolescents living with HIV.

#### Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s12978-021-01090-2.

Additional file 1: Dataset.

#### Acknowledgements

We thank Andrea Howard, Director of the T32 fellowship program for her support and encouragement during the manuscript development process from conception to the final daft. Our gratitude to Fatima Tsiouris for her input in conceiving the study, to Chloe Teasdale for all of her input in conceiving the study and in development of the manuscript, and our deep appreciation to Yael Hirsch-Moverman for editing the final version of the manuscript.

#### Authors' contributions

JN conducted, analyzed, and wrote the draft manuscript; MAC interpreted and edited the draft manuscript; WR Conceived, designed, assisted in analyzing, interpreting, and edited the manuscript. All authors read and approved the final manuscript.

#### Funding

JN was supported by the National Institute of Allergy and Infectious Diseases of the National Institutes of Health under Award Number T32Al114398. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

#### Availability of data and materials

The dataset supporting the conclusions of this article is included within the article under Additional files (see Additional file 1).

#### Ethics approval and consent to participate

The institutional review board at the US Centers for Disease Control and Prevention (CDC) provided non-research determination for this analysis. CDC Protocol number: CGH HSR protocol number 2017-519.

#### **Consent for publication**

The data in our analysis were completely de-identified and anonymous and waived by the CDC IRB for the need of consent for publication from participants.

#### **Competing interests**

None of the authors declared any financial, general or institutional competing interests. All authors had full access to the study data and take full responsibility for the accuracy of the data analyzed.

#### Author details

<sup>1</sup> ICAP At Columbia University's Mailman School of Public Health, 722 W. 168th St., New York, NY 10032, USA. <sup>2</sup> Department of Epidemiology, Mailman School of Public Health, Columbia University, New York, NY, USA. <sup>3</sup> Division of Infectious Diseases, Columbia University Irving Medical Center, New York, NY, USA.

#### Received: 23 April 2020 Accepted: 27 January 2021 Published online: 03 May 2021

#### References

- 1. Chersich MF, Newbatt E. UNICEF's contribution to the adoption and implementation of option B+ for preventing mother-to-child transmission of HIV: a policy analysis. Global Health. 2018;14(1):55.
- Muller AD, Bode S, Myer L, Roux P, von Steinbuchel N. Electronic measurement of adherence to pediatric antiretroviral therapy in South Africa. Pediatr Infect Dis J. 2008;27(3):257–62.
- Dellar RC, Dlamini S, Karim QA. Adolescent girls and young women: key populations for HIV epidemic control. J Int AIDS Soc. 2015;18(2 Suppl 1):19408–19408.
- Pettifor A, Filiatreau L, Delany-Moretlwe S. Time to strengthen HIV treatment and prevention for youth. Lancet HIV. 2019;6(11):e727–8.
- UNICEF: Children and AIDS: statistical update. https://data.unicef.org (2017). Accessed 02 Feb 2020.
- Khalifa A, Stover J, Mahy M, Idele P, Porth T, Lwamba C. Demographic change and HIV epidemic projections to 2050 for adolescents and young people aged 15–24. Glob Health Action. 2019;12(1):1662685–1662685.
- UNAIDS: Start Free, Stay Free, AIDS Free: 2019 report. https://www. unaids.org (2019). Accessed 18 Dec 2019.
- UNFPA: Adolescent pregnancy: a review of the evidence. https://www. unfpa.org (2013). Accessed 13 Dec 2019.
- Gunn JK, Asaolu IO, Center KE, Gibson SJ, Wightman P, Ezeanolue EE, Ehiri JE. Antenatal care and uptake of HIV testing among pregnant women in sub-Saharan Africa: a cross-sectional study. J Int AIDS Soc. 2016;19(1):20605.
- Musarandega R, Machekano R, Chideme M, Muchuchuti C, Mushavi A, Mahomva A, Guay L. PMTCT service uptake among adolescents and adult women attending antenatal care in selected health facilities in Zimbabwe. J Acq Immune Defic Syndr. 2017;75(2):148–55.
- Clouse K, Pettifor A, Shearer K, Maskew M, Bassett J, Larson B, Van Rie A, Sanne I, Fox MP. Loss to follow-up before and after delivery among women testing HIV positive during pregnancy in Johannesburg, South Africa. Trop Med Int Health. 2013;18(4):451–60.

- Nuwagaba-Biribonwoha H, Kiragga AN, Yiannoutsos CT, Musick BS, Wools-Kaloustian KK, Ayaya S, Wolf H, Lugina E, Ssali J, Abrams EJ. Adolescent pregnancy at antiretroviral therapy (ART) initiation: a critical barrier to retention on ART. J Int AIDS Soc. 2018;21(9):e25178.
- 13. UNAIDS: UNAIDS DATA 2020. In.: Joint United Nations Programme on HIV/AIDS Geneva; 2020.
- Swaziland H. incidence measurement survey 2: Population-based HIV impact assessment, SHIMS2 2016–2017; Summary sheet preliminary findings. Ministry of Health Swaziland: Mbabane, Swaziland; 2017.
- 15. EPHI P. ICAP. ETHIOPIA POPULATION-BASED HIV IMPACT ASSESSMENT EPHIA 2017–2018. SUMMARY SHEET: PRELIMINARY FINDINGS. In.
- Nutor JJ, Duodu PA, Agbadi P, Duah HO, Oladimeji KE, Gondwe KW. Predictors of high HIV+ prevalence in Mozambique: a complex samples logistic regression modeling and spatial mapping approaches. PLoS ONE. 2020;15(6):e0234034–e0234034.
- 17. Organization WH: Service delivery approaches to HIV testing and conselling (HTC): a strategic HTC programme framework. 2012.
- Huang X, Liu X, Chen J, Bao Y, Hou J, Lu X, Xia W, Xia H, Song A, Liu Z, et al. Evaluation of blood-based antibody rapid testing for hiv early therapy: a meta-analysis of the evidence. Front Immunol. 2018;9:1458–1458.
- Govender K, Masebo WGB, Nyamaruze P, Cowden RG, Schunter BT, Bains A. HIV prevention in adolescents and young people in the eastern and southern African region: a review of key challenges impeding actions for an effective response. Open AIDS J. 2018;12:53–67.
- Ronen K, McGrath CJ, Langat AC, Kinuthia J, Omolo D, Singa B, Katana AK, Ng'Ang AL, John-Stewart G. Gaps in adolescent engagement in antenatal care and prevention of mother-to-child HIV Transmission Services in Kenya. J Acq Immune Defic Syndr. 2017;74(1):30–7.
- Woldesenbet S, Jackson D, Lombard C, Dinh TH, Puren A, Sherman G, Ramokolo V, Doherty T, Mogashoa M, Bhardwaj S, et al. Missed opportunities along the prevention of mother-to-child transmission services cascade in South Africa: uptake, determinants, and attributable risk (the SAPMTCTE). PLoS ONE. 2015;10(7):e0132425.
- 22. Fatti G, Shaikh N, Eley B, Jackson D, Grimwood A. Adolescent and young pregnant women at increased risk of mother-to-child transmission of HIV and poorer maternal and infant health outcomes: A cohort study at public facilities in the Nelson Mandela Bay Metropolitan district, Eastern Cape. South Africa South Afr Med J. 2014;104(12):874–80.
- Teasdale CA, Alwar T, Chege D, Fayorsey R, Hawken MP, Abrams EJ. Impact of youth and adolescent friendly services on retention of 10–24-year-olds in HIV care and treatment programs in Nyanza, Kenya. J Acq Immune Defic Syndr. 2016;71(2):e56–9.
- 24. UNICEF: Adolescent HIV Prevention: turning the tide against AIDS will require more concentrated focus on adolescents and young people. https://data.unicef.org (2019). Accessed 02 Feb 2020.
- Maskew M, Bor J, MacLeod W, Carmona S, Sherman GG, Fox MP. Adolescent HIV treatment in South Africa's national HIV programme: a retrospective cohort study. Lancet HIV. 2019;6(11):e760–8.
- 26. Birdthistle I, Schaffnit SB, Kwaro D, Shahmanesh M, Ziraba A, Kabiru CW, Phillips-Howard P, Chimbindi N, Ondeng'e K, Gourlay A, Cowan FM. Evaluating the impact of the DREAMS partnership to reduce HIV incidence among adolescent girls and young women in four settings: a study protocol. BMC Public Health. 2018;18(1):1–5.
- Chimbindi N, Birdthistle I, Shahmanesh M, Osindo J, Mushati P, Ondeng'e K, Zuma T, Chiyaka T, Kyegombe N, Hargreaves J, et al. Translating DREAMS into practice: early lessons from implementation in six settings. PLoS ONE. 2018;13(12):e0208243.
- Karim QA, Baxter C, Birx D. Prevention of HIV in adolescent girls and young women: key to an AIDS-free generation. JAIDS J Acqu Immune Defic Syndr. 2017;75:S17–26.
- Forhan SE, Modi S, Houston JC, Broyles LN. Moving toward test and start: learning from the experience of universal antiretroviral therapy programs for HIV-infected pregnant/ breastfeeding women. AIDS. 2017;31(10):1489–93.
- 30. Fokam J, Sosso SM, Yagai B, Billong SC, Djubgang Mbadie RE, Kamgaing Simo R, Edimo SV, Nka AD, Tiga Ayissi A, Yimga JF, et al. Viral suppression in adults, adolescents and children receiving antiretroviral therapy in Cameroon: adolescents at high risk of virological failure in the era of "test and treat." AIDS Res Therapy. 2019;16(1):36–36.

- Rosenberg NE, Bhushan NL, Vansia D, Phanga T, Maseko B, Nthani T, Libale C, Bamuya C, Kamtsendero L, Kachigamba A, et al. Comparing youth-friendly health services to the standard of care through "Girl Power-Malawi": a quasi-experimental cohort study. J Acq Immune Defic Syndr. 2018;79(4):458–66.
- Reif LK, McNairy ML, Lamb MR, Fayorsey R, Elul B. Youth-friendly services and differentiated models of care are needed to improve outcomes for young people living with HIV. Curr Opin HIV AIDS. 2018;13(3):249–56.
- CDC: Adolescents and HIV: responding to the unique needs of teenagers in resource-limited settings. https://www.cdc.gov (2018). Accessed 2 Dec 2019.
- Mustapha M, Musiime V, Bakeera-Kitaka S, Rujumba J, Nabukeera-Barungi N. Utilization of "prevention of mother-to-child transmission" of HIV services by adolescent and young mothers in Mulago Hospital, Uganda. BMC Infect Dis. 2018;18(1):566–566.

### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

